Author’s response to reviews

Title: Salivary pellets induce a pro-inflammatory response involving the TLR4 - NF-κB pathway in gingival fibroblasts

Authors:

Heinz-Dieter Müller (heinz-dieter.mueller@zmk.unibe.ch)
Barbara Cvikl (barbara.cvikl@zmk.unibe.ch)
Adrian Lussi (adrian.lussi@zmk.unibe.ch)
Reinhard Gruber (reinhard.gruber@meduniwien.ac.at)

Version: 3 Date: 20 Jun 2016

Author’s response to reviews:

Dear Editor,

Please find enclosed the comments with our responses. We would like to thank for the valuable comments, which have been carefully addressed.

Author’s response to the comments

Editor’s comments:

1. We cannot currently accept the contributions of author "AL" as qualification for authorship, based on the statement provided. Response to editor’s comment 1: We thank the editor for the important comment. We now updated the information of the contribution of the Author Adrian Lussi (AL).

2. Please change the "Material and Methods" heading to "Methods". Response to editor’s comment 2: We thankfully follow the suggestion and adopted the "Material and Methods" heading to "Methods".
3. Please change the "Introduction" heading to "Background". Response to editor`s comment 3: Accordingly to the editor`s comment we changed the "Introduction" heading to "Background".

4. Please provide the email address of all authors. This should be placed in the title page. Response to editor`s comment 4: We adopted our manuscript and added the email addresses of all authors.

5. At this stage, please upload a single final, clean version of the manuscript. Response to editor`s comment 5:

We uploaded a clean version of the manuscript.